Author's response to reviews

Title: Association Between Tobacco Use of Pulmonologists and Implementation of Comprehensive versus Regular Smoking Cessation Counseling, in Turkey: a cross-sectional study

Authors:

   Pinar Pazarli Bostan (pinarpazarli@yahoo.com)
   Canan Karaman Demir (drcanankaraman@yahoo.com)
   Osman Elbek (osmanelbek@yahoo.com)
   Sule Akcay (msuleakcay@gmail.com)

Version: 5 Date: 24 February 2015

Author's response to reviews:

The manuscript revised to include line and page numbers.
We provided an abstract formatted according to the guidelines for authors.
We Updated ethic statements to include the name of ethics committee.
We stated in methods about written informed consent for participation.
We labelled a Conclusion section